Author’s response to reviews

Title: Changes in risk perceptions during the 2014 Ebola virus disease epidemic: results of two consecutive surveys among the general population in Lower Saxony, Germany

Authors:

Julie Obenauer (obenauermotl@etsu.edu)
Nicole Rübsamen (nicole_ruebsamen@web.de)
Ekaterine Garsevanidze (e.garsevanidze@gmail.com)
André Karch (andre.karch@helmholtz-hzi.de)
Rafael Mikolajczyk (rafael.mikolajczyk@uk-halle.de)

Version: 2 Date: 03 May 2018

Author’s response to reviews:

1. We now clarify that minors (under 16 years old) were included in the sample in the methods section of our manuscript. We now clarify in the Ethics approval and consent to participate section that consent was obtained from a parent or guardian on behalf of any participants under the age of 16.

2. We now only include one title page in our manuscript.

3. We now removed our ‘Authors response letter’ from your submission’s supplementary files.

4. We now upload our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.